



# **A Review on Distribution System State Estimation Algorithms**

Maria Fotopoulou <sup>1</sup>,\*<sup>1</sup>, Stefanos Petridis <sup>1</sup>, Ioannis Karachalios <sup>2</sup> and Dimitrios Rakopoulos <sup>1</sup>

- <sup>1</sup> Centre for Research and Technology Hellas, Chemical Process and Energy Resources Institute, 15125 Athens, Greece
- <sup>2</sup> Department of Computer Engineering and Informatics Graduate, University of Patras, 26504 Patras, Greece
- \* Correspondence: fotopoulou@certh.gr; Tel.: +30-69445-24255

**Abstract:** The modern energy requirements and the orientation towards Renewable Energy Sources (RES) integration promote the transition of distribution grids from passive, unidirectional, fossil fuel-based into active, bidirectional, environmental-friendly architectures. For this purpose, advanced control algorithms and optimization processes are implemented, the performance of which relies on the Distribution System State Estimation (DSSE). DSSE algorithms provide the Distribution System Operator (DSO) with detailed information regarding the network's state in order to derive the optimal decisions. However, this task is quite complex as the distribution system has inherent unbalance issues, often faces lack of adequate measurements, etc. The purpose of this paper is to review the DSSE algorithms that a system can incorporate with emphasis on their particular requirements, the mathematical formulation of the problem, the analysis of the existing model-based and data-driven approaches and the recommended solutions regarding observability issues, bad data detection, and meter placement strategies. Furthermore, special attention is paid to DSSE applications, including the use cases where they can be deployed, the approaches that are usually followed, the integrated distributed power supply units, as well as their future trends and challenges, thus highlighting their business-related aspects.



Citation: Fotopoulou, M.; Petridis, S.; Karachalios, I.; Rakopoulos, D. A Review on Distribution System State Estimation Algorithms. *Appl. Sci.* 2022, *12*, 11073. https://doi.org/ 10.3390/app122111073

Academic Editors: Nunzio Cennamo, Marco Carminati and Graziella Scandurra

Received: 28 September 2022 Accepted: 26 October 2022 Published: 1 November 2022

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). **Keywords:** distribution system state estimation (DSSE); smart grid; unbalanced grid; weighted least square (WLS) method; data-driven approaches; observability; pseudo-measurements; bad data; meter placement; applications

# 1. Introduction

Over the past few decades, in order to support the ongoing energy transition, the concept of the electrical network has gradually changed [1]. More specifically, for the purpose of sustainability and the enhancement of autonomy, Renewable Energy Sources (RES), storage systems and other environmental-friendly technologies have emerged, the implementation of which has transformed both Medium Voltage (MV) and Low Voltage (LV) distribution systems [2,3]. Hence, passive, fuel-based distribution is challenged by more active, viable, and decentralized concepts [4].

In order to support the operation of modern distribution systems, the Distribution System Operator (DSO) utilizes a variety of algorithms that provide the optimal results and decisions [5]. Naturally, in many cases, the input of the aforementioned algorithms relies heavily on the accurate and detailed monitoring of the system's status [6]. Consequently, an intermediate process between the acquisition of the system's available measurements and the use of control and decision-support algorithms is required. This process, i.e., Distribution System State Estimation (DSSE), constitutes an asset of great importance for the DSO as it can be used for multiple purposes, including loss monitoring, load and RES profile creation, detection of voltage limit violations and asymmetry, fault localization, outage handling, etc., as presented in Figure 1 [7].



Figure 1. The value of DSSE for the DSO.

Overall, DSSE has evolved into a research field that has gathered the interest of many experts. More specifically, accurate and efficient DSSE is considered to be a complex task, not only due to the inherent unbalances of MV and LV grids, but also due to the lack of measurements, which may cause observability issues, as well as the possible existence of bad data (either caused by damaged/inaccurate metering instruments or by cyber-attacks) that need to be identified and excluded from the estimation, as presented in Figure 2 [8]. The aforementioned challenges may be approached in a variety of ways, usually through model-based or through more recently developed, data-driven strategies, providing a pool of multifarious solutions regarding DSSE implementation [9].



Figure 2. Basic concept of DSSE implementation.

In literature, the expertise and trends related to some of the main aspects of DSSE algorithms are analyzed. For example, the authors of [10] review the requirements of state estimation and the differences between state estimation in transmission and distribution systems. Furthermore, in [11], the authors focus on model-based and forecast aided DSSE algorithms, while in [12] the authors review DSSE algorithms with special attention to data-driven approaches. Regarding the particular requirements of DSSE, in [13], DSSE algorithms are reviewed with a focus on observability issues and the proposed solutions, and in [14], DSSE algorithms and False Data Injection (FDI) attacks are analyzed. Further, from another view point, the authors of [15] focus on pilot applications of DSSE.

The purpose of this paper is to present a comprehensive review and guidelines regarding the existing DSSE algorithms, including the recommended solutions for observability, bad data detection, and meter placement issues, as well as to highlight their applications with an emphasis on the technical and business-related aspects. Therefore, it provides a more holistic approach, including the mathematical/methodological but also the practical aspects of DSSE, thus expanding the purpose of [10–15]. The rest of the paper is organized as follows: Section 2 presents the DSSE challenging attributes, Section 3 reviews the state vector options as well as the model-based, forecast aided, and data-driven DSSE algorithms, Section 4 focuses on the distribution system's observability, bad data detection, and meter placement strategies, Section 5 presents the technical requirements and applications of DSSE, Section 6 presents the future trends and challenges and, finally, Section 7 summarizes the conclusions.

#### 2. DSSE Special Attributes

Distribution systems have a number of special attributes that render their detailed representation more complex than transmission systems. Firstly, distribution systems are unbalanced by nature, as presented in Figure 3. This is a result of the unbalanced consumption as well as the existence of mutual impendences between the lines. It should be noted that in some parts of the grid, phases may be missing by design, according to the requirements and geographical location of the loads, thereby enhancing the occurring unbalances. Thus, the modelling of the components, including lines, transformers, loads, etc., as well as the power flow equations, need to be adjusted. For example, the impendence matrix of a distribution line is formulated as presented in (1), where Z is the impendence [16]. Furthermore, the active and reactive power flow between two nodes is formulated as presented in (2) and (3), respectively, and the power injection is formulated as presented in (4) and (5), where  $P_{i,j}^{ph}$  and  $Q_{i,j}^{ph}$  are the active and reactive power, respectively, flowing in phase ph between nodes i and j,  $P_i^{ph}$  and  $Q_i^{ph}$  are the active and reactive power injection, respectively, in phase ph at node i, V is the voltage magnitude and  $\delta$  is the voltage angle, G and B refer to the real and imaginary parts of the admittance matrix, respectively, l is the index referring to each of the three phases of the line, and n refers to the number of neighboring nodes [17,18]. The complexity of these formulas lies in the separate calculation of each phase's values, as the occurring unbalances cannot be modelled via single-phase equivalent approaches [17,18].

$$Z = \begin{bmatrix} Z_{aa} & Z_{ab} & Z_{ac} \\ Z_{ba} & Z_{bb} & Z_{bc} \\ Z_{ca} & Z_{cb} & Z_{cc} \end{bmatrix}.$$
 (1)

$$P_{i,j}^{ph} = V_{i}^{ph} \sum_{l=1}^{3} V_{i}^{l} [G_{i,j}^{ph,l} \cos\left(\delta_{i}^{ph} - \delta_{i}^{l}\right) + B_{i,j}^{ph,l} \sin\left(\delta_{i}^{ph} - \delta_{i}^{l}\right)] - V_{i}^{ph} \sum_{l=1}^{3} V_{j}^{l} [G_{i,j}^{ph,l} \cos\left(\delta_{i}^{ph} - \delta_{j}^{l}\right) + B_{i,j}^{ph,l} \sin\left(\delta_{i}^{ph} - \delta_{j}^{l}\right)]$$
(2)

$$Q_{i,j}^{ph} = -V_{i}^{ph} \sum_{l=1}^{3} V_{i}^{l} [G_{i,j}^{ph,l} \sin\left(\delta_{i}^{ph} - \delta_{i}^{l}\right) - B_{i,j}^{ph,l} \cos\left(\delta_{i}^{ph} - \delta_{i}^{l}\right)] - V_{i}^{ph} \sum_{l=1}^{3} V_{j}^{l} [G_{i,j}^{ph,l} \sin\left(\delta_{i}^{ph} - \delta_{j}^{l}\right) - B_{i,j}^{ph,l} \cos\left(\delta_{i}^{ph} - \delta_{j}^{l}\right)]$$
(3)

$$P_{i}^{ph} = V_{i}^{ph} \sum_{l=1}^{3} \sum_{j=1}^{n} V_{j}^{l} [G_{i,j}^{ph,l} \cos\left(\delta_{i}^{ph} - \delta_{j}^{l}\right) + B_{i,j}^{ph,l} \sin\left(\delta_{i}^{ph} - \delta_{j}^{l}\right)]$$
(4)

$$Q_{i}^{ph} = V_{i}^{ph} \sum_{l=1}^{3} \sum_{j=1}^{n} V_{j}^{l} [G_{i,j}^{ph,l} \sin\left(\delta_{i}^{ph} - \delta_{j}^{l}\right) - B_{i,j}^{ph,l} \cos\left(\delta_{i}^{ph} - \delta_{j}^{l}\right)]$$
(5)



Figure 3. Unbalanced phases in the distribution system.

Another issue that the distribution systems face is the limited availability of real-time data as a result of the sparse meter placement. Therefore, the accuracy or even the convergence of the DSSE might be negatively affected. This is an issue that can be tackled either with the placement of more instruments, which provide the DSSE with more accuracy but have a certain cost, or with the use of pseudo-measurements [19]. Pseudo-measurements are used to augment the available real measurements and are usually calculated using short-term forecasts or historical data. As a result, they are not as accurate as an actual measurement, however they require no further actions or cost from the side of the DSO [20]. A more detailed analysis regarding the methods to obtain pseudo-measurements is presented in sub-Section 4.1.

An additional factor that adds complexity to the DSSE is the configuration of the system. First and foremost, a modern, universal DSSE needs to be adaptable and equally efficient in all sorts of configurations–not just radial (which is the conventional use case) [21]. For example, the integration of RES and various innovative architectures in the MV and LV levels has promoted the research and development on ring and/or meshed configurations. Moreover, for reasons related to the increase of the system's reliability, interconnected configurations (with more than one feeder) are studied. Other than the type of configuration, the status of the switches might sometimes be vague to the DSO, as they are not always monitored. In distribution networks, switches are mostly used in order to isolate a fault or mitigate congestion issues in the distribution system. A close switch of phase ph between nodes i and j is modeled as presented in (6)–(7), and an open switch is modeled as presented in (8)–(9) [17].

$$V_i^{\rm ph} - V_i^{\rm ph} = 0 \tag{6}$$

$$\delta_i^{ph} - \delta_i^{ph} = 0 \tag{7}$$

$$P_{i,j}^{\rm ph} = 0 \tag{8}$$

$$Q_{i,i}^{ph} = 0 \tag{9}$$

Taking the aforementioned attributes into account, it is evident that the DSSE constitutes a challenging, complex task with multiple proposed approaches, which will be presented in the following Sections.

# 3. DSSE Fundamentals and Main Algorithms

The purpose of this Section is to analyze the foundations of DSSE, from the state vector options up to the main DSSE algorithms.

#### 3.1. State Vectors

The state vector of the system is defined as a set of variables, which, unlike transmission systems, may be either Node-Voltage (NV)-based, i.e., NV-DSSE, or Branch-Current (BC)-based, i.e., BC-DSSE [13]. As there is no standard process, the selection of the appropriate state vector can be determined by the criteria presented in Table 1, which compares the two state vector options in terms of applicability, complexity, efficiency, and compatibility.

NV-DSSE includes the voltages of each node and can be formed using either polar or rectangular coordinates. These state vectors are the most common in literature and also constitute the default state vectors of transmission systems [10,22]. Their main advantage, aside from their wide implementation and rich literature analysis, is that they can be used for all sorts of configurations [23], such as radial, mesh, etc. In addition, this sort of state vector is recommended for the easy incorporation of voltage measurements. However, it is noted that NV-DSSE is highly sensitive to measurement weights and is likely to face convergence issues [12]. Work related to NV-DSSE can be found in [17,24,25] using polar coordinates and in [23,26] using rectangular coordinates.

BC-DSSE includes the currents of each branch and is usually formed using rectangular coordinates but can be also formed using polar coordinates [27,28]. For these state vectors, power measurements are expressed in terms of branch currents, and their use is recommended as a more straightforward process, especially when it comes to current measurements [9,11]. Even though BC-DSSE is mostly suitable for radial distribution networks, which is a major limitation, it should be mentioned that it is considered to have simple implementation, lower computational time, lower sensitivity (to measurement weights, etc.), and higher possibility to converge, compared to NV-DSSE [11,13]. Therefore, this approach is in many cases preferred to its counterpart. Research related to BC-DSSE is presented in [27,29] using rectangular coordinates and in [28,30] using polar coordinates.

	NV-DSSE	BC-DSSE
Configurations	All (radial, meshed, etc.) [23]	Mostly limited to radial
Coordinates	Polar or rectangular [17,23–26]	Mostly rectangular [27–30]
Implementation	More complex	Relatively easy [9]
Sensitivity (weight variations, etc.) [11]	Higher (might affect convergence)	Lower
Computational time [9,11]	Higher	Lower
Compatibility with transmission system state estimation	Yes [22]	No

Table 1. Comparison between NV-DSSE and BC-DSSE [9,11–13,22–30].

# 3.2. DSSE Algorithms

The DSSE algorithms can be categorized as model-based, forecasting-aided, and datadriven. The main mathematical formulas, advantages, disadvantages, and related research are presented below.

#### 3.2.1. Conventional, Model-Based Algorithms

The most common approach regarding the core of DSSE is the Weighted Least Square (WLS) algorithm. This is a model-based solution, denoting that the details of the distribution

network need to be known to the operator beforehand. The purpose of WLS is to minimize the weighted residuals between the estimated and measured values, subjected to the distribution network's constraints. Provided that the residual vector **r** is calculated with (10), where **z** is the measurement vector, **x** is the state vector, and  $\mathbf{h}(\mathbf{x})$  is the measurement function calculated upon **x**, the objective function of the WLS is presented in (11). In the objective function, **W** is the weight matrix that denotes the operator's confidence in the measured data. It should be noted that the size of **z** is (m × 1) where m refers to the number of measurements, the size of **x** is (n × 1) where n refers to the number of states, and the size of **W** is (m × m). Obviously, m can only be lower than (or equal to) n.

$$\mathbf{r} = \mathbf{z} - \mathbf{h}(\mathbf{x}) \tag{10}$$

$$\min \mathbf{F} = \mathbf{r}^{\mathrm{T}} \mathbf{W} \mathbf{r} \tag{11}$$

The aforementioned method is characterized by its simplicity and wide use, e.g., implementation in the work of [31–36]. However, it is known to be sensitive to bad data, noise of measured values, etc. For this purpose, variations of WLS have been developed, improving its robustness at the cost of simplicity and computational resources. For example, the Least Absolute Value (LAV) constitutes a common alternative. According to this approach, the purpose is to minimize the sum of absolute values of measurement residuals, as presented in (12). Detailed implementation can be found in the work of [37,38]. This approach can be further enhanced by adding weights to each residual. This is also known as the Weighted Least Absolute Value (WLAV). The respective objective function is presented in (13), where  $w_i$  denotes the weight of the i-th residual. Research regarding WLAV can be found in [39].

$$\min \mathbf{F} = \sum_{i=1}^{m} |\mathbf{r}_i| \tag{12}$$

$$minF = \sum_{i=1}^{m} w_i |r_i|$$
(13)

Another advanced and robust model-based approach is the Least Trimmed Squares (LTS). In this case, the squared values of the residuals are calculated and ordered from the lowest to the highest. The objective function aims to select a total number, u, of lowest values and minimize their sum, as presented in (14). Related work can be found in [40].

$$\min F = \sum_{i=1}^{u} r_i^2 \tag{14}$$

Furthermore, the Least Median of Squares (LMS) method has been used for DSSE by the authors of [41]. In this case, the aim of the objective function (15) is to minimize the median–instead of the sum–of the squared, weighted residuals, where  $r_w^2$  is the matrix of squared, weighted residuals.

$$\min \mathbf{F} = \operatorname{median}(\mathbf{r}_{\mathbf{w}}^2) \tag{15}$$

The authors of [42,43] use a Generalized Maximum-likelihood (GM) approach, including normalized residuals,  $r_{S_i}$ , through the Huber cost function,  $\rho(r_{S_i})$ . The objective to be minimized is presented in (16). This approach detects and suppresses the influence of bad data and model uncertainties.

$$minF = \sum_{i=1}^{m} w_i^2 \rho(\mathbf{r}_{S_i})$$
(16)

#### 3.2.2. Forecasting-Aided Algorithms

The aforementioned conventional DSSE algorithms rely on a single set of measurements, taken on a certain moment. In this way, the evolution of the states over successive measurements is disregarded. The solution to this issue is the use of Forecasting-Aided State Estimation (FASE) [44,45]. The basic concept of FASE is to provide recursive updates of the estimated states. By these means, changes occurring during normal operation can be tracked. Moreover, since FASE is based on forecasts by nature, the usual problem of missing measurements can be addressed with the use of the forecasted states.

In these sort of algorithms, Kalman-based filters are adopted to capture the dynamics of the distribution system. A typical dynamic model is presented in (17), where k is the time instant,  $\mathbf{F}(k)$  is the is (n × n) state transition matrix,  $\mathbf{g}(k)$  denotes the trend behavior of the state trajectory, with size equal to (n × 1), and  $\mathbf{w}(k)$ , the size of which is also equal to (n × 1), models the noise, which is usually assumed to follow a Gaussian distribution with zero mean [44].

$$\mathbf{x}(\mathbf{k}+1) = \mathbf{F}(\mathbf{k})\mathbf{x}(\mathbf{k}) + \mathbf{g}(\mathbf{k}) + \mathbf{w}(\mathbf{k})$$
(17)

There is a variety of DSSE algorithms based on variations of the Kalman filter, as reported in [46–48]. In nonlinear systems, two commonly used variations are the Extended Kalman Filter (EKF) and the Unscented Kalman Filter (UKF). More specifically, the EKF is based on the linearization of nonlinear systems via the Taylor series. This approach has been adopted by the authors of [49–52] for a variety of purposes, from simple, passive distribution systems up to systems including Photovoltaics (PVs), Electric Vehicles (EVs), etc. On the other hand, the UKF is a nonlinear filtration algorithm that uses Unscented Transformation (UT) as an alternative to the linearization of nonlinear equations with the use of the Taylor series. Therefore, unlike EKF, no explicit calculation of Jacobian or Hessian matrixes are necessary in order to implement this algorithm [53]. The UFK has been adopted in [52,53].

#### 3.2.3. Data-Driven Algorithms

Due to the ascending amalgamation of Information Technologies (IT) in the energy sector, data-driven algorithms have gained popularity in the field of state estimation over the past few years [54]. A major advantage of data-driven algorithms is that they assist in overcoming the issues that are encountered in other DSSE algorithms, which depend on parameters and models of distribution networks, are complicated, time-consuming, and very sensitive to initial conditions. Nevertheless, it should be highlighted that for this purpose a significant amount of data are required [55,56].

In literature, there is a variety of Artificial Intelligence (AI) and Machine Learning (ML) approaches including: (i) supervised, (ii) unsupervised, and (iii) reinforcement learning [57] that are also utilized in several fields of study [58,59]. In supervised learning, a labeled dataset, including input data and their corresponding output, is required. In the training phase of the algorithm, the relation between the provided inputs and outputs is defined as a function. Artificial Neural Networks (ANN), the popularity of which is increased, fall in this sort of learning. In addition, it should be highlighted that a well-known subcategory of ANN are the Deep Neural Networks (DNN), which in contrast to simple ANN have more than one hidden layer. On the other hand, in unsupervised learning, there are no complete and clean-labeled datasets. This is a sort of learning that finds previously unknown patterns in datasets and discovers the output. Finally, reinforcement learning is based on the interaction with the environment. In this case, an agent interacts with the environment by performing actions and learns either from errors or rewards [60].

In literature, the most common data-driven DSSE algorithms are based on ANN. For example, the authors of [61,62] use ANN algorithms to perform DSSE in MV and LV levels. Studies and experimental tests show that once the ANN is trained, it is characterized by low computational complexity. Since the estimation process is computationally simple, a main advantage is that it can be executed on a low-cost hardware. Of course, the accuracy depends on the number and quality of the available measurements. Other developers choose to add more than one hidden layer to their ANN-based solutions, i.e., DNN. For example, the authors of [63] have developed a DNN framework to perform DSSE, which relies on a deep Recurrent Neural Network (RNN) postulating module. Additionally, following a more complex approach, in [64–66] physics-aware neural networks are proposed. In this

case, the underlying physical model, meaning the structure of the distribution network, is exploited, unlike the previous learning models, where the physics of the underlying distribution network are overlooked. By pruning the unneeded neural network connections, the overfitting behavior of the algorithm is prohibited. Furthermore, according to the respective research, physics-aware neural networks are often compared to WLS in order to highlight the superiority of their performance.

Of course, there are also developers who prefer to combine data-driven and modelbased approaches, resulting in hybrid DSSE algorithms [67,68]. Indicatively, the authors of [67] have developed a hybrid algorithm that combines a DNN state estimator and a WLAV state estimator in order to ensure robust results at a low time scale. In this case, except from the DNN, the methodology includes a random forest topology identification, which is also a data-driven approach.

# 3.2.4. Summary and Comparison between DSSE Algorithms

All the aforementioned DSSE algorithms are presented in Figure 4 and summarized in Table 2. It is highlighted that the WLS constitutes the most widely implemented solution. Yet, its model-based nature is directly linked with the accurate knowledge necessity of the distribution system's parameters, the dependency on the quality of the data, etc. This issue can be solved mostly through modern, data-based algorithms such as ANN, the ascending implementation of which challenges the dominance of their tested-in-time, model-based counterparts.



Figure 4. Main sorts of DSSE algorithms.

Table 2. DSSE algorithms [31–65].

Method	Description	Advantages and Disadvantages
WLS [31-36]	Minimization of the weighted residuals between the estimated and measured values. Objective function (11).	Common, simple, fast model-based approach, but sensitive to data.
LAV [37,38]	Minimization of the sum of absolute values of measurement residuals. Objective function (12).	Model-based approaches. Mostly variations of
WLAV [39]	Variation of LAV including weights. Objective function (13).	WLS, trying to overcome the issue of sensitivity to bad data and model parameters. Enhanced
LTS [40]	Utilization of selected number of residuals in the objective function (14).	robustness but also limited implementation in literature and higher resources requirements such as computational cost, memory usage, etc.
LMS [41]	Use of the median in the objective function (15).	Information regarding the architecture and
GM [42,43]	Generalized maximum-likelihood approach using the Huber cost function in (16).	parameters of the distribution system is a necessity.

# Table 2. Cont.

Method	Description	Advantages and Disadvantages
Kalman filter [46–48]	FASE, providing recursive updates of the estimated states with the use of Kalman filter and dynamic models such as (17).	Changes in normal operation can be tracked. Further, since FASE is based on forecasts by nature, missing measurements can be addressed with the use of the forecasted states. However, this sort of algorithms is more complex and not as widely used as WLS.
EKF [49-52]	FASE, extended variation of Kalman filter with the use of Taylor series for linearization of nonlinear systems.	Widely used variation of Kalman filter for linearization of nonlinear systems. Yet, not as widely used as WLS.
UKF [52,53]	FASE, variation of Kalman filter with the use of UT for linearization of nonlinear systems.	Variation of Kalman filter for linearization of nonlinear systems where no explicit calculation of Jacobian or Hessian matrices is necessary. Limited use of this approach in literature.
ANN [61,62]	Purely data-driven DSSE approach including three layers, trained with data provided by the DSO. Modern, data-driven approach, whe	
DNN [63]	Data-driven approach, sub-category of ANN, with more hidden layers.	initial conditions, etc. does not affect the result. However, an adequate database is required.
Physics-aware neural network [64–66]	Data-driven approach where the neural network takes into account the structure of the distribution network.	Advanced, data-driven approach where the physics of the network are not overlooked, thus prohibiting the overfitting behavior of the algorithm. Yet, in this case too, an adequate database is required.
Hybrid [67,68]	Various possibilities, e.g., [67]: DNN-based DSSE, supported by random forest algorithm for topology identification. WLAV is used to ensure the robustness of the estimation.	Combination of the advantages of various sorts of algorithms at the cost of having a less simple methodology.

#### 4. Auxiliary Algorithms

The main auxiliary algorithms that support the DSSE concern: (i) observability, (ii) bad data, and (iii) meter placement. The purpose of this Section is to review these algorithms.

#### 4.1. Observability

The limited availability of real time data may challenge the observability of the distribution system. In order to provide the DSSE with enough data to converge and produce accurate results, pseudo-measurements are used [69]. Pseudo-measurements can be generated in a variety of ways. In literature, two main categories are distinguished, i.e., (i) probabilistic and statistical approaches and (ii) learning-based approaches [9,13], as presented in Table 3.

Following probabilistic and statistical approaches, the authors of [70–72] have used Gaussian Mixture Models (GMM) and Expectation Maximization (EM). The Gaussian mixture,  $f(z|\gamma)$ , comprises a weighted, finite sum of Gaussian probability density functions,  $f(z|\mu_i, \Sigma_i)$ , as presented in (18), where  $M_c$  is the number of mixture components. EM is used in order to obtain the parameters of the mixture components, i.e., weights, means, and variances, as presented in [73].

$$f(z|\gamma) = \sum_{i=1}^{M_c} w_i f(z|\mu_i, \Sigma_i)$$
(18)

Another way to obtain pseudo-measurements that falls into the category of probabilistic and statistical approaches is presented in [72] and involves the calculation of correlation coefficients between measurements obtained from the main substation and non-monitored electrical quantities and the application of regression analysis. The authors also compare the results provided by the correlation approach to the respective ones provided by the GMM.

When it comes to learning-based approaches, the popularity of DNN seems to be increased. For example, the authors of [74] have developed a DNN that produces pseudomeasurements, which are used as input in a typical WLS DSSE. It is reported that the computation time for the training of the DNN is approximately 10 min, using 17,520 sets of real power flow measurements and injections. Furthermore, the authors of [75] have developed a Parallel Distribution Processing network (PDP) for the generation of pseudomeasurements related to the load time-series of MV distribution networks. In a more complex manner, the authors of [76] propose the implementation of a nonlinear autoregressive exogenous (NARX) model for the efficient forecasting/estimation of loads.

Table 3. Auxiliary algorithms for the acquisition of pseudo-measurements [70–72,74–76].

Method	Description	Advantages and Disadvantages
GMM and EM [70-72]	Probabilistic and statistical approaches using Gaussian distributions, correlations, etc.	Mature algorithms, easy implementation, but with increased sensitivity and difficulties when it comes
Correlation coefficients [72]		to large systems.
DNN [74] PDP [75] NARX [76]	Solutions with learning-based approaches, using mostly neural networks.	Modern, advanced algorithms, but with database requirements.

## 4.2. Bad Data Detection

Bad data detection is of substantial importance for successfully estimating the system's true state. Bad data can stem from: (i) erroneous measuring data [77], (ii) system faults [78], and (iii) False Data Injection Attacks (FDIAs) [79–82]. Thus, the discovery of false data can also help DSOs identify possible attacks in their system.

The algorithms utilized for performing bad data detection can be classified into two main categories: (i) model-based detection algorithms and (ii) data-driven algorithms [83], as presented in Table 4.

Model-based detection algorithms are prediction methods that measure the similarity between the predicted states and the actual field measurements. In literature, model-based bad data detection algorithms are used extensively. In [84–89] the authors use the L<sub>2</sub>-norm that is the Euclidean distance of the residual and compare it to a certain threshold as presented in (19), where measurement  $z_i$  is considered faulty when its Euclidean distance from its respective calculation upon the predicted state,  $h(x_i)$ , is greater than the threshold, e.

$$\mathbf{P}_{L_2}(\mathbf{z}_i) = \begin{cases} 1, \text{ if } \|\mathbf{z}_i - \mathbf{h}(\mathbf{x}_i)\|_2 > e \\ 0, \text{ otherwise} \end{cases}$$
(19)

Similar to the previous one, the authors of [90–94] utilize the Largest Normalized Residual (LNR) test that is presented in (20), where,  $\sigma_E = \text{diag}(\mathbf{E})$ , is the residual error covariance matrix.

$$\mathbf{P}_{\text{LNR}}(\mathbf{z}_{i}) = \begin{cases} 1, \text{ if } \|\mathbf{z}_{i} - \mathbf{h}(\mathbf{x}_{i}) / \sigma_{\text{E}}\|_{\infty} > e \\ 0, & \text{otherwise} \end{cases}$$
(20)

Last but not least, in [78,95–97] the authors present the Chi-square test for the purpose of FDIA detection. Chi-squared is given by (21), where J(x) is the cost function of the DSSE.

$$\mathbf{P}_{\chi^2}(\mathbf{z}_i) = \begin{cases} 1, & \text{if } J(\mathbf{x}_i) > e \\ 0, & \text{otherwise} \end{cases}$$
(21)

In certain applications,  $L_2$ -norm test outperformed the Chi-square test [78]. Finally, by combining the  $L_2$ -norm and the LNR tests, the authors of [98] achieved better results in FDIA detection than the cases of using only one detection test.

Unfortunately, it was mathematically proven by [79] that the model-based detection algorithms discussed above are vulnerable to attack vectors inserted from individuals that have prior knowledge of the system. These vectors are, in some cases, undetectable from the tests presented.

Data-driven methods for bad data detection can solve this issue. Supervised learning techniques, in which the ML models are trained with labeled data, are used extensively in literature. In more detail, the authors of [99] use Linear Regression (LR) to fit a measurement vector in the linear model. If the vector does not fit in the model, FDIA is detected. Another technique used is the Support Vector Machine (SVM), which is the most common ML technique for bad data detection. The authors of [96,100–103] utilize SVM classifiers for bad data and FDIA detection. Although they are extensively used, in most publications, SVMs play the role of the baseline algorithm for identifying whether the proposed ML method of the publication has a better performance. In [100], an ANN with one hidden layer that can automatically infer underlying physical relationships exploiting sensor data is shown. The authors of [104] train an ANN for forecasting current and voltage measurements and identifying whether FDIA occurs in the system. In [82], the RNNs' ability to exhibit temporal dynamic behavior, finding patterns in sequential data, is examined. Here the RNNs are utilized as sequence classifiers to detect potential measurement manipulation. The authors of [103] extend this work for detecting FDIAs in a subset of the system's nodes. More recently, the authors of [105] exploit the Convolutional Neural Networks' (CNNs) excellence in pattern recognition in FDIA detection. Similar to CNNs, another ML algorithm commonly used in image processing is employed in [102] for detecting FDIAs, called the Margin Setting Algorithm (MSA). MSA is compared with the SVM and ANN algorithms. Other supervised learning algorithms found in literature are the k-nearest neighbors (k-NN) and the extended nearest neighbor (ENN) algorithms [106].

Unsupervised learning, in which unlabeled data are provided for classification, is another ML technique used in bad data detection. In [107], a k-means clustering (KMC) algorithm is used. KMC's objective is to separate the unlabeled data into k different clusters. Another method similar to KMC is proposed in [108,109] and called the Fuzzy Clustering (FC) algorithm. The difference between FC and KMC is that in FC the boundaries of the clusters are overlapping, so a single sample could belong in multiple clusters.

Overall, data-driven algorithms for bad data detection are proven to be more accurate than their model-based counterparts. Although the model-based algorithms are simpler and more robust in most of the cases, their accuracy is a considerable tradeoff.

Method	Description	Advantages and Disadvantages
Model-based [78,84–98]	L <sub>2</sub> -norm [84–89], LNR [90–94], Chi-square [78,95–97], Combination [98]	Robust and simple methods. Not as accurate.
Data-driven [82,96,99–109]	LR [99], SVM [96,100–103], ANN [100,104], RNN [82,103], CNN [105], MSA [102], k-NN [106], ENN [106], KMC [107], FC [108,109]	Very accurate methods. Need large amounts of data.

Table 4. Bad data detection methods [78,82,84–109].

# 4.3. Meter Placement

The meter placement in a distribution system constitutes a key decision problem. For this purpose, three main sorts of algorithms are distinguished: (i) rule-based, (ii) metaheuristic, and (iii) optimization with an objective function subjected to a set of constraints [110]. In more detail, rule-based algorithms comprise of a number of rules that lead to the easy and fast solution of a problem at the cost of providing non-optimal solutions. Metaheuristic algorithms are usually bio-inspired and more evolved than rule-based algorithms. Indicative examples are Particle Swarm Optimization (PSO), Tabu Search (TS), etc. [111]. By using these sort of algorithms, sufficiently good solutions can be obtained but global optimality is not guaranteed. However, the most recent trends indicate the use of optimizers, which

aim to maximize/minimize an objective function, limited by constraints, in order to obtain the optimal solution [112]. The main idea is to model constraints such as energy balances, power flows, voltage limitations, etc., and create a space for possible/feasible decisions. The purpose is to find the optimal set of decisions that maximizes/minimizes the value of the objective function. These problems can be Mixed Integer Linear Programming (MILP), Mixed Integer Nonlinear Programming (MINLP), etc., depending on the nature of the problem [113].

In literature, all of the aforementioned strategies have been recruited for meter placement, as presented in Table 5. For example, a rule-based meter placement strategy is presented in [21]. The developed methodology is simple (but not optimal) and is applicable only to radial distribution networks. Its purpose is to reduce the overall meter cost. Metaheuristics have been used by the authors of [114–117]. More specifically, ref. [114] uses a Genetic Algorithm (GA) in order to ensure adequate accuracy at the lowest possible cost. In [115], binary PSO is implemented in order for the maximization of state estimation accuracy, limited by a maximum number of meters. Further, the authors of [116,117] use the fruit fly and TS metaheuristic algorithms, respectively, for the minimization of measurement units that can sustain observability. Regarding optimization approaches, ref. [19] proposes the use of Mixed Integer Semidefinite Programming (MISDP), which is a subfield of convex optimization, where the objective is the maximization of state estimation accuracy. The same objective is used in [118], approached with an optimization non-relaxed, Boolean convex model. The authors of [25] propose MISDP for the minimization of worst-case estimation errors.

Tab	le 5.	Meter	placement strategies	[19,21,25,114–118]
-----	-------	-------	----------------------	--------------------

Method	Description	Advantages and Disadvantages
Rule-based [21]Set of rules for meter placement with low cost.		Easy and fast algorithms, but not optimal.
Metaheuristic [114–117]	GA [114], PSO [115], fruit fly [116], TS [117], etc. for: (i) maximization of accuracy or, (ii) minimization of units while sustaining observability, etc.	Efficient bio-inspired algorithms which do not guarantee global optimal solutions.
Optimization [19,25,118]	MISDP, etc., usually for the maximization of accuracy.	Optimal, advanced algorithms, more demanding than the others.

It is evident that optimization algorithms constitute the most advanced approaches, while rule-based algorithms constitute the simplest and least optimal ones. This is a typical trade-off between more modern and more conventional approaches. Nevertheless, it is at the operator's discretion to choose the one that suits their purposes the best.

# 5. Technical Requirements and Applications

This Section is related to the practical and business-related aspects of DSSE, including technical requirements, potential applications, pilots, etc. The purpose is to showcase the components installed on the system, the computational resources, the applicability of the aforementioned algorithms, and the approaches that are usually followed in reality.

# 5.1. Technical Requirements

It is evident that the most important requirement for a DSSE algorithm is the availability of measurements. Installing metering devices on a system is a necessary step for monitoring its performance and achieve the desired observability [119]. For having an accurate and up to date representation of the network, DSSE needs to be applied at least once for every hour of the day, so new measurements are needed for every new application of the algorithm. For example, the authors of [120] apply the DSSE in 10 min intervals, and in [121] the algorithm is performed every 30 min. The most common measuring components found in distribution systems are Phasor Measurement Units (PMUs) and Smart Meters (SM) [122–124]. PMUs offer synchronous power, voltage, and current measurements, as well as phase angles. SMs, on the other hand, record information such as energy consumption, voltage levels, current, and power factor. In specific distribution systems where only SMs exist and, consequently, no information concerning the phase angles, DSSE algorithms have to make assumptions about phase angle computation [125–127]. In addition, PMUs produce measurement samples in a higher frequency (reaching 30 kHz) than SMs (0.277 mHz–16.7 mHz) [13]. One can conclude that PMUs provide more sophisticated information than SMs. However, PMUs are not vastly used in distribution systems because of the high per unit costs. This demonstrates the need for developing algorithms for optimal PMU number and placement while lowering the costs [128].

The metering devices send measurement information to the Supervisory Control and Data Acquisition (SCADA) system under the IEC 60,870 communication protocol [129]. The SCADA is a control system architecture that contains computing devices, databases, and various interfaces that enable the real time monitoring and control of a distribution system [130]. Data-driven approaches of DSSE as well as algorithms associated with DSSE, such as forecasting models for pseudo-measurement creation and bad data detection, need large amounts of historical data to be trained and function properly. Thus, databases of SCADA, capable to perform DSSE, must be able to hold years of hourly or sub-hourly power, voltage, and current data [131].

In conclusion, a lot of resources are needed to perform DSSE successfully. Performing DSSE and evaluating the distribution system's state is paramount as RES penetration and the use of electrical cars is on the rise [132,133].

# 5.2. Applications

Having discussed all the basic theory and technical requirements regarding DSSE, this sub-Section is dedicated to applications as well as actual projects where DSSE has been implemented.

Regarding the applications of DSSE in distribution systems, research is mostly focused on RES penetration and "green" technologies, due to the ongoing energy transition. In this sense, it is quite common to find studies where DSSE is performed on distribution systems with high PV penetration. For example, the authors of [134] have performed DSSE in a distribution system with PVs using a typical WLS algorithm. Similar concepts have been studied by [51,135], with the use of a variation of WLAV and a variation of EKF, respectively. Moreover, ref. [136] has performed DSSE in a distribution system that includes not only PVs but also Wind Turbines (WTs). In this case, the DSSE utilizes a DNN. In a more interesting scenario, ref. [137] has performed DSSE in a hybrid AC/DC system including PVs, WTs and diesel generators. The estimation is performed with the use of WLS, supported by a DNN. Finally, in a slightly different direction, the authors of [50] perform DSSE in a distribution system that is used to charge Electric Vehicles (EVs). For this purpose, a variation of the EKF is deployed.

The various applications and methodologies found in literature are summarized in Table 6. It is noted that all sorts of algorithms that were discussed in sub-Section 3.2 have been used, meaning that there is no unique, optimal solution when it comes to actual implementation.

Application	Approach
PV [51,134,135]	Model based, such as WLS [134], WLAV [135] but also FASE, such as EKF variations [51].
PV and WT [136]	Data-driven DSSE, with DNN [136].
PV, WT and diesel generator [137]	Model-based WLS DSSE coupled with DNN [137].
EV charging [50]	FASE DSSE, with a variation of EKF [50].

**Table 6.** Applications of DSSE [50,51,134–137].

Furthermore, there are many research and innovation projects, pilot use cases, Horizon 2020 programs, etc., where DSSE is applied on real smart grids. A map of the most wellknown cases is presented in Figure 5, indicating the sort of algorithm that was deployed. More specifically, in the context of EVOLVDSO, DSSE was implemented in a 74-node LV distribution system in Portugal and a 77-node LV distribution system in France [138]. In both cases, the overall approach was based on ANN. Furthermore, the French DSO, Enedis, applied WLS DSSE in a MV system called the VENTEEA smart grid demonstrator [120]. A WLS DSSE was also applied on a MV smart grid in the UK in the context of the Low Carbon London project [139]. The Swiss DSO applied WLS DSSE as well, in a LV 102-node distribution system located in Switzerland [140]. For the purpose of SuSTAINABLE project, a WLS DSSE was demonstrated in a MV/LV distribution system in Portugal [141]. The Slovenian Research Agency funded the development of an EKF-based DSSE in a 97-node LV distribution system in Slovenia [50]. In Denmark, a WLS DSSE supported by an ANN for the generation of pseudo-measurements was demonstrated in a MV 52-node distribution system [142]. The SmartSCADA project demonstrated a WLS DSSE in a LV distribution system in Germany [143]. The LV SCADA project developed an ANN DSSE and demonstrated it in a LV system in Portugal [144]. Finally, the largest contribution regarding actual DSSE application is the one of INTERPRETER, which is a Horizon 2020 program with three demo-sites, in Belgium, Spain, and Denmark, where WLS DSSE is implemented by the current research group [145,146].



Figure 5. DSSE pilot project implementation [50,120,138–145].

Judging by the real-life applications, it can be concluded that the main option in DSSE remains WLS. This is mostly attributed to the fact that it constitutes an easy, fast, and tested-in-time algorithm.

#### 6. Future Trends and Challenges

Taking the presented analysis into account, it is estimated that in the near future, DSSE will be implemented at a higher scale (instead of being mostly implemented in pilots) including large MV and LV distribution systems of all configurations, with a variety of DER, supporting advanced optimization algorithms that are nowadays utilized by the DSOs. In this sense, the DSOs are also expected to exploit the possible contribution of DSSE in the field of ancillary services which are derived from efficient monitoring, such as congestion management, black start, etc., thus further enhancing the cooperation between the DSOs and Transmission System Operators (TSOs) [147]. In addition, stability assessment–which is an important issue for power systems–could be related to DSSE algorithms for either stand-alone or connected systems, taking into account various aspects such as reduced order transfer function models of droop-controlled inverters via Jordan continued-fraction expansion [148,149], etc.

Furthermore, regarding the design of the DSSE, a future trend will probably be the more extended use of neural networks, which constitute a modern approach with an increasingly strong presence in the energy sector, either for state estimation per se or for pseudo-measurement generation and bad data detection [82].

However, all these benefits come with a few challenges. For example, a more standardized approach regarding the selection of state vectors, main and auxiliary algorithms, meter placement strategies, etc., needs to be established, promoting the easier implementation to future adopters, which is a major requirement for market penetration. Furthermore, it should be noted that the renovation of the distribution system, the installation of PMUs and SMs, etc., comes to a certain cost. Further, for the implementation of data-driven algorithms, large databases and advanced engineering expertise are required.

In the end, it can be concluded that, given the necessary requirements, DSSE may transform the passive, insufficiently monitored distribution system into an active, key part of the ongoing energy transition.

# 7. Conclusions

This paper presents a review of DSSE algorithms, aiming to provide an inclusive approach, from the purely theoretical up to the more business-related aspects. It highlights the usability of the DSSE and analyzes the main sorts of algorithms found in literature along with their advantages and disadvantages. In addition, it presents and compares the auxiliary algorithms that enhance the observability of the system, detect bad data, and guide the DSO towards the optimal meter placement strategies. Finally, it reviews the technical aspects and applications of DSSE, including requirements, potential use cases with enhanced RES penetration, and actual projects, thus providing a more practical analysis, paving the way towards market penetration. Overall, it is concluded that:

- WLS is the most commonly used algorithm, not only in theoretical development but also in actual applications.
- Data-driven algorithms challenge the dominance of model-based counterparts.
- DSSE can play an important role in the ongoing energy transition, but in order to do so in a large scale, standardized solutions should be established.

**Author Contributions:** Conceptualization, M.F., S.P. and D.R.; methodology, M.F., S.P. and D.R.; writing—original draft preparation, M.F., S.P. and I.K.; writing—review and editing, M.F., S.P. and D.R.; supervision, D.R.; project administration, D.R.; funding acquisition, D.R. All authors have read and agreed to the published version of the manuscript.

**Funding:** This research was funded by European Commission's H2020 programme, INTERPRETER, under grant agreement No 864360, https://cordis.europa.eu/project/id/864360 (accessed on 26 September 2022).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Acknowledgments: This review paper has been written in the context of the European Union's Horizon 2020 research and innovation programme, INTERPRETER (Interoperable tools for an efficient management and effective planning of the electricity grid).

**Conflicts of Interest:** The authors declare no conflict of interest.

# References

- 1. Mah, D.N.-Y.; Wu, Y.-Y.; Ip, J.C.-M.; Hills, P.R. The role of the state in sustainable energy transitions: A case study of large smart grid demonstration projects in Japan. *Energy Policy* **2013**, *63*, 726–737. [CrossRef]
- 2. Mahmud, N.; Zahedi, A. Review of control strategies for voltage regulation of the smart distribution network with high penetration of renewable distributed generation. *Renew. Sustain. Energy Rev.* **2016**, *64*, 582–595. [CrossRef]
- 3. Das, C.K.; Bass, O.; Kothapalli, G.; Mahmoud, T.S.; Habibi, D. Overview of energy storage systems in distribution networks: Placement, sizing, operation, and power quality. *Renew. Sustain. Energy Rev.* **2018**, *91*, 1205–1230. [CrossRef]
- 4. Anjana, K.; Shaji, R. A review on the features and technologies for energy efficiency of smart grid. *Int. J. Energy Res.* 2018, 42, 936–952. [CrossRef]
- Gust, G.; Biener, W.; Brandt, T.; Dallmer-Zerbe, K.; Neumann, D.; Wille-Haussmann, B. Decision Support for Distribution Grid Planning. In Proceedings of the 2016 IEEE International Energy Conference (ENERGYCON), Leuven, Belgium, 4–8 April 2016; pp. 1–7.
- 6. Alassery, F. Advanced metering infrastructure smart metering based on cloud architecture for low voltage distribution networks in application of smart grid monitoring. *Sustain. Comput. Inform. Syst.* **2022**, *35*, 100747. [CrossRef]
- 7. Wang, G.; Giannakis, G.; Chen, J.; Sun, J. Distribution system state estimation: An overview of recent developments. *Front. Inf. Technol. Electron. Eng.* **2019**, *20*, 4–17. [CrossRef]
- Watitwa, J.; Awodele, K. A Review on Active Distribution System State Estimation. In Proceedings of the 2019 Southern African Universities Power Engineering Conference/Robotics and Mechatronics/Pattern Recognition Association of South Africa (SAUPEC/RobMech/PRASA), Bloemfontein, South Africa, 28–30 January 2019; pp. 726–731.
- 9. Primadianto, A.; Lu, C.-N. A Review on Distribution System State Estimation. *IEEE Trans. Power Syst.* 2017, 32, 3875–3883. [CrossRef]
- Sarada Devi, M.S.N.G.; Yesuratnam, G. Comparison of State Estimation Process on Transmission and Distribution Systems. In Advances in Decision Sciences, Image Processing, Security and Computer Vision; Satapathy, S.C., Raju, K.S., Shyamala, K., Krishna, D.R., Favorskaya, M.N., Eds.; Learning and Analytics in Intelligent Systems; Springer International Publishing: Cham, Switzerland, 2020; Volume 4, pp. 414–423. ISBN 978-3-030-24317-3.
- Ahmad, F.; Rasool, A.; Ozsoy, E.; Sekar, R.; Sabanovic, A.; Elitaş, M. Distribution system state estimation-A step towards smart grid. *Renew. Sustain. Energy Rev.* 2018, *81*, 2659–2671. [CrossRef]
- 12. Radhoush, S.; Bahramipanah, M.; Nehrir, H.; Shahooei, Z. A Review on State Estimation Techniques in Active Distribution Networks: Existing Practices and Their Challenges. *Sustainability* **2022**, *14*, 2520. [CrossRef]
- 13. Dehghanpour, K.; Wang, Z.; Wang, J.; Yuan, Y.; Bu, F. A Survey on State Estimation Techniques and Challenges in Smart Distribution Systems. *IEEE Trans. Smart Grid* **2019**, *10*, 2312–2322. [CrossRef]
- 14. Bhattar, P.L.; Pindoriya, N.M.; Sharma, A. A combined survey on distribution system state estimation and false data injection in cyber-physical power distribution networks. *IET Cyber-Phys. Syst. Theory Appl.* **2021**, *6*, 41–62. [CrossRef]
- 15. Táczi, I.; Sinkovics, B.; Vokony, I.; Hartmann, B. The Challenges of Low Voltage Distribution System State Estimation—An Application Oriented Review. *Energies* **2021**, *14*, 5363. [CrossRef]
- 16. Gómez-Expósito, A.; Conejo, A.J.; Cañizares, C.A. *Electric Energy Systems: Analysis and Operation*; CRC Press: Boca Raton, FL, USA, 2016; ISBN 978-1-4200-0727-5.
- Majumdar, A.; Pal, B.C. A three-phase state estimation in unbalanced distribution networks with switch modelling. In Proceedings of the 2016 IEEE First International Conference on Control, Measurement and Instrumentation (CMI), Kolkata, India, 8–10 January 2016; pp. 474–478. [CrossRef]
- 18. Hansen, C.; Debs, A. Power system state estimation using three-phase models. *IEEE Trans. Power Syst.* **1995**, *10*, 818–824. [CrossRef]
- 19. Xygkis, T.C.; Korres, G.N. Optimized Measurement Allocation for Power Distribution Systems Using Mixed Integer SDP. *IEEE Trans. Instrum. Meas.* 2017, *66*, 2967–2976. [CrossRef]
- Clements, K.A. The Impact of Pseudo-Measurements on State Estimator Accuracy. In Proceedings of the 2011 IEEE Power and Energy Society General Meeting, Detroit, MI, USA, 24–28 July 2011; pp. 1–4.
- Baran, M.E.; Zhu, J.; Kelley, A.W. Meter Placement for Real-Time Monitoring of Distribution Feeders. In Proceedings of the Power Industry Computer Applications Conference, Salt Lake City, UT, USA, 7–12 May 1995; pp. 228–233.
- 22. Kotha, S.K.; Rajpathak, B. Power System State Estimation using Non-Iterative Weighted Least Square method based on Wide Area Measurements with maximum redundancy. *Electr. Power Syst. Res.* **2022**, 206, 107794. [CrossRef]
- 23. Gholami, M.; Tehrani-Fard, A.; Lehtonen, M.; Moeini-Aghtaie, M.; Fotuhi-Firuzabad, M. A Novel Multi-Area Distribution State Estimation Approach for Active Networks. *Energies* **2021**, *14*, 1772. [CrossRef]

- 24. Kotsonias, A.; Asprou, M.; Hadjidemetriou, L.; Kyriakides, E. State Estimation for Distribution Grids with a Single-Point Grounded Neutral Conductor. *IEEE Trans. Instrum. Meas.* 2020, 69, 8167–8177. [CrossRef]
- Yao, Y.; Liu, X.; Li, Z. Robust Measurement Placement for Distribution System State Estimation. *IEEE Trans. Sustain. Energy* 2019, 10, 364–374. [CrossRef]
- 26. Lu, C.; Teng, J.; Liu, W.-H. Distribution system state estimation. IEEE Trans. Power Syst. 1995, 10, 229–240. [CrossRef]
- 27. Melo, I.D.; Pereira, J.L.; Ribeiro, P.F.; Variz, A.M.; Oliveira, B.C. Harmonic state estimation for distribution systems based on optimization models considering daily load profiles. *Electr. Power Syst. Res.* **2019**, *170*, 303–316. [CrossRef]
- Wang, H.; Schulz, N. A Revised Branch Current-Based Distribution System State Estimation Algorithm and Meter Placement Impact. *IEEE Trans. Power Syst.* 2004, 19, 207–213. [CrossRef]
- Pau, M.; Pegoraro, P.A.; Sulis, S. Efficient Branch-Current-Based Distribution System State Estimation Including Synchronized Measurements. *IEEE Trans. Instrum. Meas.* 2013, 62, 2419–2429. [CrossRef]
- Pau, M.; Ponci, F.; Monti, A.; Sulis, S.; Muscas, C.; Pegoraro, P.A. An Efficient and Accurate Solution for Distribution System State Estimation with Multiarea Architecture. *IEEE Trans. Instrum. Meas.* 2017, 66, 910–919. [CrossRef]
- Meriem, M.; Bouchra, C.; Abdelaziz, B.; Jamal, S.O.B.; Faissal, E.M.; Nazha, C. Study of State Estimation Using Weighted-Least-Squares Method (WLS). In Proceedings of the 2016 International Conference on Electrical Sciences and Technologies in Maghreb (CISTEM), Marrakech, Morocco, 26–28 October 2016; pp. 1–5.
- Pau, M.; Pegoraro, P.A.; Sulis, S. Performance of Three-Phase WLS Distribution System State Estimation Approaches. In Proceedings of the 2015 IEEE International Workshop on Applied Measurements for Power Systems (AMPS), Aachen, Germany, 23–25 September 2015; pp. 1–6.
- 33. Shafiei, M.; Nourbakhsh, G.; Arefi, A.; Ledwich, G.; Pezeshki, H. Single Iteration Conditional Based DSSE Considering Spatial and Temporal Correlation. *Int. J. Electr. Power Energy Syst.* **2019**, *107*, 644–655. [CrossRef]
- Angioni, A.; Shang, J.; Ponci, F.; Monti, A. Real-Time Monitoring of Distribution System Based on State Estimation. *IEEE Trans. Instrum. Meas.* 2016, 65, 2234–2243. [CrossRef]
- Kumar, C.S.; Rajawat, K.; Chakrabarti, S.; Pal, B.C. Robust distribution system state estimation with hybrid measurements. *IET Gener. Transm. Distrib.* 2020, 14, 3250–3259. [CrossRef]
- Jin, T.; Shen, X. A Mixed WLS Power System State Estimation Method Integrating a Wide-Area Measurement System and SCADA Technology. *Energies* 2018, 11, 408. [CrossRef]
- 37. Gol, M.; Abur, A. LAV Based Robust State Estimation for Systems Measured by PMUs. *IEEE Trans. Smart Grid* 2014, 5, 1808–1814. [CrossRef]
- Dobakhshari, A.S.; Azizi, S.; Abdolmaleki, M.; Terzija, V. Linear LAV-based state estimation integrating hybrid SCADA/PMU measurements. *IET Gener. Transm. Distrib.* 2020, 14, 1583–1590. [CrossRef]
- 39. Celik, M.; Abur, A. A robust WLAV state estimator using transformations. IEEE Trans. Power Syst. 1992, 7, 106–113. [CrossRef]
- Weng, Y.; Negi, R.; Liu, Q.; Ilić, M.D. Robust state-estimation procedure using a Least Trimmed Squares pre-processor. In Proceedings of the ISGT 2011, Anaheim, CA, USA, 17–19 January 2011; pp. 1–6. [CrossRef]
- Mili, L.; Phaniraj, V.; Rousseeuw, P. Least median of squares estimation in power systems. *IEEE Trans. Power Syst.* 1991, 6, 511–523. [CrossRef]
- Zhao, J.; Huang, C.; Mili, L.; Zhang, Y.; Min, L. Robust Medium-Voltage Distribution System State Estimation Using Multi-Source Data. In Proceedings of the 2020 IEEE Power & Energy Society Innovative Smart Grid Technologies Conference (ISGT), Washington, DC, USA, 17–20 February 2020; pp. 1–5.
- 43. Zhao, J.; Zhang, G.; La Scala, M.; Wang, Z. Enhanced Robustness of State Estimator to Bad Data Processing Through Multiinnovation Analysis. *IEEE Trans. Ind. Inform.* 2017, 13, 1610–1619. [CrossRef]
- 44. Huang, Y.-F.; Werner, S.; Huang, J.; Kashyap, N.; Gupta, V. State Estimation in Electric Power Grids: Meeting New Challenges Presented by the Requirements of the Future Grid. *IEEE Signal Process. Mag.* **2012**, *29*, 33–43. [CrossRef]
- Liu, H.; Hu, F.; Su, J.; Wei, X.; Qin, R. Comparisons on Kalman-Filter-Based Dynamic State Estimation Algorithms of Power Systems. *IEEE Access* 2020, *8*, 51035–51043. [CrossRef]
- Shafiei, M.; Ledwich, G.; Nourbakhsh, G.; Arefi, A.; Pezeshki, H. Layered Based Augmented Complex Kalman Filter for Fast Forecasting-Aided State Estimation of Distribution Networks. *arXiv* 2018, arXiv:1804.08298.
- 47. Durgaprasad, G.; Thakur, S.S. Robust dynamic state estimation of power systems based on M-Estimation and realistic modeling of system dynamics. *IEEE Trans. Power Syst.* **1998**, *13*, 1331–1336. [CrossRef]
- 48. da Silva, A.M.L.; Do Coutto Filho, M.; de Queiroz, J.F. State forecasting in electric power systems. *IEE Proc. C Gener. Transm. Distrib.* **1983**, 130, 237–244. [CrossRef]
- 49. Zhang, Z.; Qiu, J.; Ma, W. Adaptive Extended Kalman Filter with Correntropy Loss for Robust Power System State Estimation. *Entropy* **2019**, *21*, 293. [CrossRef]
- 50. Antončič, M.; Papič, I.; Blažič, B. Robust and Fast State Estimation for Poorly-Observable Low Voltage Distribution Networks Based on the Kalman Filter Algorithm. *Energies* **2019**, *12*, 4457. [CrossRef]
- Barchi, G.; Macii, D. A photovoltaics-aided interlaced extended Kalman filter for distribution systems state estimation. Sustain. Energy Grids Netw. 2021, 26, 100438. [CrossRef]
- 52. Ćetenović, D.; Ranković, A. Optimal parameterization of Kalman filter based three-phase dynamic state estimator for active distribution networks. *Int. J. Electr. Power Energy Syst.* **2018**, 101, 472–481. [CrossRef]

- 53. Wan, E.A.; Van Der Merwe, R. The Unscented Kalman Filter for Nonlinear Estimation. In Proceedings of the IEEE 2000 Adaptive Systems for Signal Processing, Communications, and Control Symposium (Cat. No.00EX373), Lake Louise, AB, Canada, 4 October 2000; pp. 153–158.
- Simpson, R.H. Power System Data Base Management. In Proceedings of the Conference Record of 2000 Annual Pulp and Paper Industry Technical Conference (Cat. No.00CH37111), Atlanta, GA, USA, 19–23 June 2000; pp. 79–83.
- 55. Wang, Q.; Li, F.; Tang, Y.; Xu, Y. Integrating Model-Driven and Data-Driven Methods for Power System Frequency Stability Assessment and Control. *IEEE Trans. Power Syst.* **2019**, *34*, 4557–4568. [CrossRef]
- 56. Matijašević, T.; Antić, T.; Capuder, T. A systematic review of machine learning applications in the operation of smart distribution systems. *Energy Rep.* **2022**, *8*, 12379–12407. [CrossRef]
- 57. Sah, S. Machine Learning: A Review of Learning Types. Math. Comput. Sci. 2020. [CrossRef]
- Sharma, P.; Said, Z.; Kumar, A.; Nižetić, S.; Pandey, A.; Hoang, A.T.; Huang, Z.; Afzal, A.; Li, C.; Le, A.T.; et al. Recent Advances in Machine Learning Research for Nanofluid-Based Heat Transfer in Renewable Energy System. *Energy Fuels* 2022, 36, 6626–6658. [CrossRef]
- Rajamoorthy, R.; Arunachalam, G.; Kasinathan, P.; Devendiran, R.; Ahmadi, P.; Pandiyan, S.; Muthusamy, S.; Panchal, H.; Kazem, H.A.; Sharma, P. A novel intelligent transport system charging scheduling for electric vehicles using Grey Wolf Optimizer and Sail Fish Optimization algorithms. *Energy Sources Part A Recover. Util. Environ. Eff.* 2022, 44, 3555–3575. [CrossRef]
- 60. Dridi, S. Supervised Learning–A Systematic Literature Review. 2021. Available online: https://www.researchgate.net/publication/354996999\_Supervised\_Learning\_-\_A\_Systematic\_Literature\_Review (accessed on 26 September 2022).
- 61. Bernieri, A.; Betta, G.; Liguori, C.; Losi, A. Neural networks and pseudo-measurements for real-time monitoring of distribution systems. *IEEE Trans. Instrum. Meas.* **1996**, 45, 645–650. [CrossRef]
- Ferdowsi, M.; Benigni, A.; Lowen, A.; Zargar, B.; Monti, A.; Ponci, F. A Scalable Data-Driven Monitoring Approach for Distribution Systems. *IEEE Trans. Instrum. Meas.* 2015, 64, 1292–1305. [CrossRef]
- Zhang, L.; Wang, G.; Giannakis, G.B. Distribution System State Estimation Via Data-Driven and Physics-Aware Deep Neural Networks. In Proceedings of the 2019 IEEE Data Science Workshop (DSW), Minneapolis, MN, USA, 2–5 June 2019; pp. 258–262. [CrossRef]
- Zamzam, A.S.; Sidiropoulos, N.D. Physics-Aware Neural Networks for Distribution System State Estimation. *IEEE Trans. Power* Syst. 2020, 35, 4347–4356. [CrossRef]
- 65. Tran, M.-Q.; Zamzam, A.S.; Nguyen, P.H. Enhancement of Distribution System State Estimation Using Pruned Physics-Aware Neural Networks. In Proceedings of the 2021 IEEE Madrid PowerTech, Madrid, Spain, 28 June–2 July 2021. [CrossRef]
- Tran, M.-Q.; Zamzam, A.; Nguyen, P.; Pemen, G. Multi-Area Distribution System State Estimation Using Decentralized Physics-Aware Neural Networks. *Energies* 2021, 14, 3025. [CrossRef]
- 67. Huang, M.; Wei, Z.; Sun, G.; Zang, H. Hybrid State Estimation for Distribution Systems With AMI and SCADA Measurements. *IEEE Access* 2019, 7, 120350–120359. [CrossRef]
- Zamzam, A.S.; Fu, X.; Sidiropoulos, N.D. Data-Driven Learning-Based Optimization for Distribution System State Estimation. IEEE Trans. Power Syst. 2019, 34, 4796–4805. [CrossRef]
- Fantin, C.A.; Castillo, M.R.C.; de Carvalho, B.E.B.; London, J.B.A. Using Pseudo and Virtual Measurements in Distribution System State Estimation. In Proceedings of the 2014 IEEE PES Transmission & Distribution Conference and Exposition–Latin America (PES T&D-LA), Medellin, Colombia, 10–13 September 2014; pp. 1–6.
- Singh, R.; Pal, B.; Jabr, R. Distribution system state estimation through Gaussian mixture model of the load as pseudo-measurement. *IET Gener. Transm. Distrib.* 2010, 4, 50–59. [CrossRef]
- Singh, R.; Pal, B.C.; Jabr, R.A. Statistical Representation of Distribution System Loads Using Gaussian Mixture Model. *IEEE Trans.* Power Syst. 2010, 25, 29–37. [CrossRef]
- Manitsas, E.; Singh, R.; Pal, B.; Strbac, G. Modelling of Pseudo-Measurements for Distribution System State Estimation. In Proceedings of the CIRED Seminar 2008: SmartGrids for Distribution, Frankfurt, Germany, 23–24 June 2008; p. 42.
- 73. Dempster, A.P.; Laird, N.M.; Rubin, D.B. Maximum Likelihood from Incomplete Data via the EM Algorithm. J. R. Stat. Soc. Ser. B (Methodol.) 1977, 39, 1–22.
- 74. Manitsas, E.; Singh, R.; Pal, B.C.; Strbac, G. Distribution System State Estimation Using an Artificial Neural Network Approach for Pseudo Measurement Modeling. *IEEE Trans. Power Syst.* 2012, 27, 1888–1896. [CrossRef]
- 75. Wu, J.; He, Y.; Jenkins, N. A robust state estimator for medium voltage distribution networks. *IEEE Trans. Power Syst.* 2013, 28, 1008–1016. [CrossRef]
- Hayes, B.P.; Gruber, J.K.; Prodanovic, M. A Closed-Loop State Estimation Tool for MV Network Monitoring and Operation. *IEEE Trans. Smart Grid* 2015, 6, 2116–2125. [CrossRef]
- Monticelli, A.; Garcia, A. Reliable Bad Data Processing for Real-Time State Estimation. *IEEE Trans. Power Appar. Syst.* 1983, PAS-102, 1126–1139. [CrossRef]
- Manandhar, K.; Cao, X.; Hu, F.; Liu, Y. Detection of Faults and Attacks Including False Data Injection Attack in Smart Grid Using Kalman Filter. *IEEE Trans. Control Netw. Syst.* 2014, 1, 370–379. [CrossRef]
- 79. Liu, Y.; Ning, P.; Reiter, M.K. False data injection attacks against state estimation in electric power grids. *ACM Trans. Inf. Syst. Secur.* **2011**, *14*, 1–33. [CrossRef]

- Rahman, M.A.; Mohsenian-Rad, H. False Data Injection Attacks against Nonlinear State Estimation in Smart Power Grids. In Proceedings of the 2013 IEEE Power & Energy Society General Meeting, Vancouver, BC, Canada, 21–25 July 2013; pp. 1–5.
- Liang, G.; Zhao, J.; Luo, F.; Weller, S.R.; Dong, Z.Y. A Review of False Data Injection Attacks Against Modern Power Systems. *IEEE Trans. Smart Grid* 2017, *8*, 1630–1638. [CrossRef]
- Ayad, A.; Farag, H.E.Z.; Youssef, A.; El-Saadany, E.F. Detection of False Data Injection Attacks in Smart Grids Using Recurrent Neural Networks. In Proceedings of the 2018 IEEE Power & Energy Society Innovative Smart Grid Technologies Conference (ISGT), Washington, DC, USA, 19–22 February 2018; pp. 1–5.
- 83. Musleh, A.S.; Chen, G.; Dong, Z.Y. A Survey on the Detection Algorithms for False Data Injection Attacks in Smart Grids. *IEEE Trans. Smart Grid* **2020**, *11*, 2218–2234. [CrossRef]
- Kallitsis, M.G.; Bhattacharya, S.; Stoev, S.; Michailidis, G. Adaptive Statistical Detection of False Data Injection Attacks in Smart Grids. In Proceedings of the 2016 IEEE Global Conference on Signal and Information Processing (GlobalSIP), Washington, DC, USA, 7–9 December 2016; pp. 826–830.
- Khalid, H.M.; Peng, J.C.-H. Immunity Toward Data-Injection Attacks Using Multisensor Track Fusion-Based Model Prediction. IEEE Trans. Smart Grid 2017, 8, 697–707. [CrossRef]
- Xu, R.; Wang, R.; Guan, Z.; Wu, L.; Wu, J.; Du, X. Achieving Efficient Detection Against False Data Injection Attacks in Smart Grid. IEEE Access 2017, 5, 13787–13798. [CrossRef]
- Sreenath, J.G.; Meghwani, A.; Chakrabarti, S.; Rajawat, K.; Srivastava, S.C. A recursive state estimation approach to mitigate false data injection attacks in power systems. In Proceedings of the 2017 IEEE Power & Energy Society General Meeting, Chicago, IL, USA, 16–20 July 2017; pp. 1–5. [CrossRef]
- Duan, J.; Zeng, W.; Chow, M.-Y. Resilient Distributed DC Optimal Power Flow Against Data Integrity Attack. *IEEE Trans. Smart Grid* 2018, *9*, 3543–3552. [CrossRef]
- Musleh, A.S.; Khalid, H.M.; Muyeen, S.M.; Al-Durra, A. A Prediction Algorithm to Enhance Grid Resilience Toward Cyber Attacks in WAMCS Applications. *IEEE Syst. J.* 2019, 13, 710–719. [CrossRef]
- Chung, H.-M.; Li, W.-T.; Yuen, C.; Chung, W.-H.; Wen, C.-K. Local Cyber-Physical Attack with Leveraging Detection in Smart Grid. In Proceedings of the 2017 IEEE International Conference on Smart Grid Communications (SmartGridComm), Dresden, Germany, 23–27 October 2017; pp. 461–466.
- Khalaf, M.; Youssef, A.; El-Saadany, E. Detection of False Data Injection in Automatic Generation Control Systems Using Kalman Filter. In Proceedings of the 2017 IEEE Electrical Power and Energy Conference (EPEC), Saskatoon, SK, Canada, 22–25 October 2017; pp. 1–6.
- Karimipour, H.; Dinavahi, V. On False Data Injection Attack against Dynamic State Estimation on Smart Power Grids. In Proceedings of the 2017 IEEE International Conference on Smart Energy Grid Engineering (SEGE), Oshawa, ON, Canada, 14–17 August 2017; pp. 388–393.
- 93. Karimipour, H.; Dinavahi, V. Robust Massively Parallel Dynamic State Estimation of Power Systems Against Cyber-Attack. *IEEE Access* 2018, *6*, 2984–2995. [CrossRef]
- Shi, W.; Wang, Y.; Jin, Q.; Ma, J. PDL: An Efficient Prediction-Based False Data Injection Attack Detection and Location in Smart Grid. In Proceedings of the 2018 IEEE 42nd Annual Computer Software and Applications Conference (COMPSAC), Tokyo, Japan, 23–27 July 2018; pp. 676–681.
- 95. Liu, T.; Sun, Y.; Liu, Y.; Gui, Y.; Zhao, Y.; Wang, D.; Shen, C. Abnormal traffic-indexed state estimation: A cyber–physical fusion approach for Smart Grid attack detection. *Futur. Gener. Comput. Syst.* **2015**, *49*, 94–103. [CrossRef]
- Chen, P.-Y.; Yang, S.; McCann, J.A.; Lin, J.; Yang, X. Detection of false data injection attacks in smart-grid systems. *IEEE Commun. Mag.* 2015, 53, 206–213. [CrossRef]
- Jiang, Y.; Hui, Q. Kalman Filter with Diffusion Strategies for Detecting Power Grid False Data Injection Attacks. In Proceedings of the 2017 IEEE International Conference on Electro Information Technology (EIT), Lincoln, NE, USA, 14–17 May 2017; pp. 254–259.
- Zhao, J.; Zhang, G.; La Scala, M.; Dong, Z.Y.; Chen, C.; Wang, J. Short-Term State Forecasting-Aided Method for Detection of Smart Grid General False Data Injection Attacks. *IEEE Trans. Smart Grid* 2017, *8*, 1580–1590. [CrossRef]
- 99. Yip, S.-C.; Wong, K.; Hew, W.-P.; Gan, M.-T.; Phan, R.C.-W.; Tan, S.-W. Detection of energy theft and defective smart meters in smart grids using linear regression. *Int. J. Electr. Power Energy Syst.* 2017, *91*, 230–240. [CrossRef]
- Ferragut, E.M.; Laska, J.; Olama, M.M.; Ozmen, O. Real-Time Cyber-Physical False Data Attack Detection in Smart Grids Using Neural Networks. In Proceedings of the 2017 International Conference on Computational Science and Computational Intelligence (CSCI), Las Vegas, NV, USA, 14–16 December 2017; pp. 1–6.
- Esmalifalak, M.; Liu, L.; Nguyen, N.; Zheng, R.; Han, Z. Detecting Stealthy False Data Injection Using Machine Learning in Smart Grid. *IEEE Syst. J.* 2017, 11, 1644–1652. [CrossRef]
- Wang, Y.; Amin, M.M.; Fu, J.; Moussa, H.B. A Novel Data Analytical Approach for False Data Injection Cyber-Physical Attack Mitigation in Smart Grids. *IEEE Access* 2017, 5, 26022–26033. [CrossRef]
- Binna, S.; Kuppannagari, S.R.; Engel, D.; Prasanna, V.K. Subset Level Detection of False Data Injection Attacks in Smart Grids. In Proceedings of the 2018 IEEE Conference on Technologies for Sustainability (SusTech), Long Beach, CA, USA, 11–13 November 2018; pp. 1–7.

- 104. El Hariri, M.; Youssef, T.A.; Habib, H.F.; Mohammed, O. Online false data detection and lost packet forecasting system using time series neural networks for IEC 61850 sampled measured values. In Proceedings of the 2017 IEEE Power & Energy Society Innovative Smart Grid Technologies Conference (ISGT), Washington, DC, USA, 23–26 April 2017; pp. 1–5. [CrossRef]
- 105. Wang, D.; Wang, X.; Zhang, Y.; Jin, L. Detection of power grid disturbances and cyber-attacks based on machine learning. *J. Inf. Secur. Appl.* **2019**, *46*, 42–52. [CrossRef]
- 106. Yan, J.; Tang, B.; He, H. Detection of False Data Attacks in Smart Grid with Supervised Learning. In Proceedings of the 2016 International Joint Conference on Neural Networks (IJCNN), Vancouver, BC, Canada, 24–29 July 2016; pp. 1395–1402.
- McLaughlin, S.; Holbert, B.; Fawaz, A.; Berthier, R.; Zonouz, S. A Multi-Sensor Energy Theft Detection Framework for Advanced Metering Infrastructures. *IEEE J. Sel. Areas Commun.* 2013, 31, 1319–1330. [CrossRef]
- 108. Viegas, J.L.; Vieira, S.M. Clustering-Based Novelty Detection to Uncover Electricity Theft. In Proceedings of the 2017 IEEE International Conference on Fuzzy Systems (FUZZ-IEEE), Naples, Italy, 9–12 July 2017; pp. 1–6.
- Zanetti, M.; Jamhour, E.; Pellenz, M.; Penna, M.; Zambenedetti, V.; Chueiri, I. A Tunable Fraud Detection System for Advanced Metering Infrastructure Using Short-Lived Patterns. *IEEE Trans. Smart Grid* 2019, 10, 830–840. [CrossRef]
- Camacho, E.F.; Bordons, C. Model Predictive Control: Advanced Textbooks in Control and Signal Processing, 2nd ed; Springer: London, UK; New York, NY, USA, 2004; ISBN 978-0-85729-398-5.
- Abdel-Basset, M.; Abdel-Fatah, L.; Sangaiah, A.K. Metaheuristic Algorithms: A Comprehensive Review. In *Computational Intelligence for Multimedia Big Data on the Cloud with Engineering Applications*; Elsevier: Amsterdam, The Netherlands, 2018; pp. 185–231. ISBN 978-0-12-813314-9.
- 112. Pereira, J.L.J.; Oliver, G.A.; Francisco, M.B.; Cunha, S.S.; Gomes, G.F. A Review of Multi-objective Optimization: Methods and Algorithms in Mechanical Engineering Problems. *Arch. Comput. Methods Eng.* **2022**, *29*, 2285–2308. [CrossRef]
- 113. Klanšek, U. A comparison between MILP and MINLP approaches to optimal solution of nonlinear discrete transportation problem. *Transport* **2014**, *30*, 135–144. [CrossRef]
- 114. Liu, J.; Tang, J.; Ponci, F.; Monti, A.; Muscas, C.; Pegoraro, P.A. Trade-Offs in PMU Deployment for State Estimation in Active Distribution Grids. *IEEE Trans. Smart Grid* **2012**, *3*, 915–924. [CrossRef]
- 115. Raposo, A.A.; Rodrigues, A.B.; da Silva, M.D.G. Optimal meter placement algorithm for state estimation in power distribution networks. *Electr. Power Syst. Res.* 2017, 147, 22–30. [CrossRef]
- 116. Ramachandran, B.; Bellarmine, G.T. Improving observability using optimal placement of phasor measurement units. *Int. J. Electr. Power Energy Syst.* **2014**, *56*, 55–63. [CrossRef]
- 117. Peng, J.; Sun, Y.; Wang, H. Optimal PMU placement for full network observability using Tabu search algorithm. *Int. J. Electr. Power Energy Syst.* 2006, 28, 223–231. [CrossRef]
- Xygkis, T.C.; Korres, G.N.; Manousakis, N.M. Fisher Information-Based Meter Placement in Distribution Grids via the D-Optimal Experimental Design. *IEEE Trans. Smart Grid* 2018, 9, 1452–1461. [CrossRef]
- 119. El-Hawary, M.E. The Smart Grid-State-of-the-art and Future Trends. Electr. Power Compon. Syst. 2014, 42, 239–250. [CrossRef]
- 120. Croteau, D.; Carre, O. Distribution state estimation: Outcomes from a field implementation aimed at tackling MV mastering in the presence of distributed energy resources (DER). *CIRED–Open Access Proc. J.* **2017**, 2017, 1715–1717. [CrossRef]
- 121. Al-Wakeel, A.; Wu, J.; Jenkins, N. State estimation of medium voltage distribution networks using smart meter measurements. *Appl. Energy* **2016**, *184*, 207–218. [CrossRef]
- 122. Pegoraro, P.A.; Tang, J.; Liu, J.; Ponci, F.; Monti, A.; Muscas, C. PMU and Smart Metering Deployment for State Estimation in Active Distribution Grids. In Proceedings of the 2012 IEEE International Energy Conference and Exhibition (ENERGYCON), Florence, Italy, 9–12 September 2012; pp. 873–878.
- 123. Heydt, G.T. The Next Generation of Power Distribution Systems. IEEE Trans. Smart Grid 2010, 1, 225–235. [CrossRef]
- 124. Albu, M.; Heydt, G.T.; Cosmescu, S.-C. Versatile platforms for wide area synchronous measurements in power distribution systems. In Proceedings of the North American Power Symposium 2010, Arlington, TX, USA, 26–28 September 2010. [CrossRef]
- 125. Deka, D.; Backhaus, S.; Chertkov, M. Structure Learning in Power Distribution Networks. *IEEE Trans. Control. Netw. Syst.* 2015, 5, 1061–1074. [CrossRef]
- 126. Peppanen, J.; Reno, M.J.; Broderick, R.J.; Grijalva, S. Distribution System Model Calibration With Big Data From AMI and PV Inverters. *IEEE Trans. Smart Grid* 2016, 7, 2497–2506. [CrossRef]
- 127. Weng, Y.; Liao, Y.; Rajagopal, R. Distributed Energy Resources Topology Identification via Graphical Modeling. *IEEE Trans. Power* Syst. 2017, 32, 2682–2694. [CrossRef]
- Carta, D.; Benigni, A.; Muscas, C. Model Order Reduction for PMU-Based State Estimation in Distribution Grids. *IEEE Syst. J.* 2018, 12, 2711–2720. [CrossRef]
- 129. SCADA Communication and Protocols. Available online: https://instrumentationtools.com/scada-communication-and-protocols/ (accessed on 26 September 2022).
- 130. Tomsovic, K.; Bakken, D.; Venkatasubramanian, V.; Bose, A. Designing the Next Generation of Real-Time Control, Communication, and Computations for Large Power Systems. *Proc. IEEE* 2005, *93*, 965–979. [CrossRef]
- 131. Hippert, H.S.; Pedreira, C.E.; Souza, R.C. Neural networks for short-term load forecasting: A review and evaluation. *IEEE Trans. Power Syst.* **2001**, *16*, 44–55. [CrossRef]
- 132. Fotopoulou, M.; Rakopoulos, D.; Blanas, O. Day Ahead Optimal Dispatch Schedule in a Smart Grid Containing Distributed Energy Resources and Electric Vehicles. *Sensors* **2021**, *21*, 7295. [CrossRef]

- Petridis, S.; Blanas, O.; Rakopoulos, D.; Stergiopoulos, F.; Nikolopoulos, N.; Voutetakis, S. An Efficient Backward/Forward Sweep Algorithm for Power Flow Analysis through a Novel Tree-Like Structure for Unbalanced Distribution Networks. *Energies* 2021, 14, 897. [CrossRef]
- Bandara, W.G.C.; Almeida, D.; Godaliyadda, R.I.; Ekanayake, M.P.; Ekanayake, J. A complete state estimation algorithm for a three-phase four-wire low voltage distribution system with high penetration of solar PV. *Int. J. Electr. Power Energy Syst.* 2021, 124, 106332. [CrossRef]
- 135. Song, S.; Xiong, H.; Lin, Y.; Huang, M.; Wei, Z.; Fang, Z. Robust three-phase state estimation for PV-Integrated unbalanced distribution systems. *Appl. Energy* **2022**, 322, 119427. [CrossRef]
- 136. Menke, J.-H.; Bornhorst, N.; Braun, M. Distribution system monitoring for smart power grids with distributed generation using artificial neural networks. *Int. J. Electr. Power Energy Syst.* 2019, 113, 472–480. [CrossRef]
- 137. Huang, M.; Zhao, J.; Wei, Z.; Pau, M.; Sun, G. Decentralized robust state estimation for hybrid AC/DC distribution systems with smart meters. *Int. J. Electr. Power Energy Syst.* 2022, 136, 107656. [CrossRef]
- 138. Teixeira, H.; Pereira Barbeiro, P.N.; Pereira, J.; Bessa, R.; Matos, P.G.; Lemos, D.; Morais, A.C.; Caujolle, M.; Sebastian-Viana, M. A State Estimator for LV Networks: Results from the EvolvDSO Project. In Proceedings of the CIRED Workshop 2016, Helsinki, Finland, 14–15 June 2016; p. 136.
- Dragovic, J.; Pudjianto, D.; Djiapic, P.; Bilton, M.; Strbac, G. Application of Distribution System State Estimation on Engineering Instrumentation Zones of Low Carbon London. In Proceedings of the 23rd International Conference on Electricity Distribution (CIRED 2015), Lyon, France, 15–18 June 2015.
- 140. Eggenschwiler, L.; Favre-Perrod, P.; Nauts, O.; Rampazzo, J.; Mousavi, O. Performance evaluation of distribution system state estimator using different measurement devices. In Proceedings of the CIRED 2019, Madrid, Spain, 3–6 June 2019.
- 141. Sustainable Project. Advanced Local Distribution Grid Monitoring/State-Estimation; Sustainable Project: Porto, Portugal; Available online: https://repositorio-aberto.up.pt/bitstream/10216/83284/2/44855.pdf (accessed on 20 September 2022).
- 142. Pokhrel, B.R.; Bak-Jensen, B.; Pillai, J.R. Integrated Approach for Network Observability and State Estimation in Active Distribution Grid. *Energies* **2019**, *12*, 2230. [CrossRef]
- 143. Waeresch, D.; Brandalik, R.; Wellssow, W.H.; Jordan, J.; Bischler, R.; Schneider, N. Field test of a linear three-phase low-voltage state estimation system based on smart meter data. *CIRED–Open Access Proc. J.* **2017**, 2017, 1773–1776. [CrossRef]
- 144. Barbeiro, P.; Pereira, J.; Teixeira, H.; Seca, L.; Silva, P.; Silva, N.; Melo, F. LV SCADA Project: In-Field Validation of a Distribution State Estimation Tool for LV Networks. In Proceedings of the CIRED Workshop 2016, Helsinki, Finland, 14–15 June 2016; p. 134.
- 145. Petridis, S.; Blanas, O.; Rakopoulos, D.; Stergiopoulos, F.; Nikolopoulos, N.; Voutetakis, S.; Andritsakis, E.; Tagkoulis, N. *LV Network Topology and Parameter Estimation with High Data Availability*; Centre for Research and Technology Hellas (CERTH): Thessaloniki, Greece, 2020.
- 146. Petridis, S.; Bintoudi, A.D.; Veliskaki, A.; Karapatsias, V.; Fotopoulou, M.; Rakopoulos, D.; Ioannidis, D.; Tzovaras, D. State Estimation in Unbalanced Smart Grids. In Proceedings of the 2022 International Conference on Smart Energy Systems and Technologies (SEST), Eindhoven, The Netherlands, 5–7 September 2022; pp. 1–6.
- 147. Manousakis, N.M.; Korres, G.N. Application of State Estimation in Distribution Systems with Embedded Microgrids. *Energies* **2021**, *14*, 7933. [CrossRef]
- 148. Rui, W.; Qiuye, S.; Pinjia, Z.; Yonghao, G.; Dehao, Q.; Peng, W. Reduced-Order Transfer Function Model of the Droop-Controlled Inverter via Jordan Continued-Fraction Expansion. *IEEE Trans. Energy Convers.* **2020**, *35*, 1585–1595. [CrossRef]
- Wang, R.; Sun, Q.; Qin, D.; Li, Y.; Li, X.; Wang, P. Steady-state Stability Assessment of AC-busbar Plug-in Electric Vehicle Charging Station with Photovoltaic. J. Mod. Power Syst. Clean Energy 2020, 8, 884–894. [CrossRef]